Author's response to reviews

Title: Space-time clustering of childhood central nervous system tumours in Yorkshire, UK

Authors:

Richard JQ McNally (richard.mcnally@ncl.ac.uk)
Peter W James (peter.james@ncl.ac.uk)
Susan V Picton (susan.picton@leedsth.nhs.uk)
Patricia A McKinney (p.a.mckinney@leeds.ac.uk)
Marlous van Laar (m.vanlaar@leeds.ac.uk)
Richard G Feltbower (r.g.feltbower@leeds.ac.uk)

Version: 5 Date: 10 January 2012

Author's response to reviews:

Typographical error in the page numbers for reference 7. This has now been corrected to "129-136".